Author’s response to reviews

Title: The potential value of serum pepsinogen for the diagnosis of atrophic gastritis among the health check-up populations in China: a diagnostic clinical research

Authors:
Yuling Tong (irenetyl@sina.com; 961459033@qq.com)
Yulian Wu (hbtranslation@sina.com)
Zhenya Song (89533384@qq.com)
Yingying Yu (41250433@qq.com)
Xinyan Yu (yuxinyan1214@163.com)

Version: 5 Date: 18 Jun 2017

Author’s response to reviews:

Manuscript ID: BMGE-D-16-00541 (Research Article)

Title: The potential value of serum pepsinogen for the diagnosis of atrophic gastritis among the health check-up populations in China: a diagnostic clinical research

Journal: BMC Gastroenterology

Response to Reviewers’ comments

Dear Dr. Megan Haran

We should apologize that there was some confusion regarding the previous comments. we are very grateful for your detailed explanations for the comments to our manuscript entitled “The potential value of serum pepsinogen for the diagnosis of atrophic gastritis among the health check-up populations in China: a diagnostic clinical research”.

We amend the dataset according to the comments.

We hope that this new very will be suitable for publication in your renowned journal. Please do not hesitate to contact me if you need further information regarding this manuscript.
Best wishes,

Yulian Wu
General Surgery Department, The Second Affiliated Hospital of Zhejiang University, School of Medicine, NO.88 Jiefang Road, Hangzhou, Zhejiang Province, P.R. China
Tel: +86-571-89713988
E-mail: hbtranslation@sina.com

Zhenya Song
International HealthCare Center, The Second Affiliated Hospital of Zhejiang University, School of Medicine, NO.88 Jiefang Road, Hangzhou, Zhejiang Province, P.R. China
Tel: +86-571-89713988
E-mail: 89533384@qq.com